Author’s response to reviews

Title: Dual Contraception Method Utilization and Associated Factors among Sexually Active Women on Antiretroviral Therapy in Gondar City, northwest, Ethiopia: a cross sectional study

Authors:

Fewuze Abay (fewuzeabay@gmail.com)

Hedija Yeshita (hedijayeshash@yahoo.com)

Fantahun Mekonnen (fantahunaye@gmail.com)

Mekonnen Sisay (mekudesu@gmail.com)

Version: 2 Date: 20 Jan 2020

Author’s response to reviews:

Comments were addressed.